Author’s response to reviews

Title: Architecture of the Short External Rotator Muscles of the Hip

Authors:

Kevin Parvaresh (kcparvaresh@gmail.com)
Charles Chang (charles.chang@yale.edu)
Ankur Patel (ankeel@gmail.com)
Richard Lieber (rlieber@sralab.org)
Scott Ball (stball@mail.ucsd.edu)
Samuel Ward (s1ward@ucsd.edu)

Version: 2 Date: 06 Dec 2019

Author’s response to reviews:

Technical Comments:

Editor Comments:

The authors have revised the manuscript according to reviewers’ comments in most aspects.

They should discuss their choice of total resection instead of diagnostic biopsy.

My sincere apology for the incorrect term. As English is not our first language, our term was incorrect. It should be diagnostic biopsy. It has been corrected. (Abstract section, line 33, page 2; Case presentations section, line 126, page 7)

Thanks for your frank and sincere suggestion.

There are many repetitions. the manuscript should be summarized in a more fluent style.

Thanks for your suggestion. It has been summarized and shortened. (Discussion and Conclusion section, line 152/ 162/ 176-178/ 198/ 206/ 233, page 8-12)
The English language should be checked by a native speaker.

Yes, it has been checked and revised by a native speaker.

Thanks for your suggestion.